Author's response to reviews

Title: Effectiveness of a school-based physical activity intervention on obesity in school children: a nonrandomized controlled trial

Authors:

Xiao-Hui Li (happyhuihui778@163.com)
Shenting Lin (shentinglin@163.com)
Hongxia Guo (cpcdcghx@126.com)
Yanli Huang (hylxwd@126.com)
Lijing Wu (wulj0519@163.com)
Zilong Zhang (josephchangzzl@gmail.com)
Jun Ma (majunt@bjmu.edu.cn)
Hai-Jun Wang (whjun1@bjmu.edu.cn)

Version: 2 Date: 26 August 2014

Author's response to reviews:

Dear reviewer:
The following changes have been made to my revised version:
1. Line numbers and page numbers have been added.
2. Section heading 'Introduction' and 'Disclosure' have been changed into 'Background' and 'Competing Interests' respectively.
3. The Funding statement has been included in the 'Acknowledge' section.

Best regards
Hai-Jun Wang